Author's response to reviews

Title: The development and evaluation of the Australian Child and Adolescent Recommended Food Score: a cross-sectional study

Authors:

Skye Marshall (Skye.Huxley@uon.edu.au)
Jane F Watson (Jane.Watson@newcastle.edu.au)
Tracy L Burrows (Tracy.Burrows@newcastle.edu.au)
Maya Guest (Maya.Guest@newcastle.edu.au)
Clare E Collins (clare.collins@newcastle.edu.au)

Version: 6 Date: 1 November 2012

Author's response to reviews: see over
1st November, 2012

Dear Editor

RE: MS 7221291926993588: The development and evaluation of the Australian Child and Adolescent Recommended Food Score: a cross-sectional study

Thank you for the opportunity to submit a revised manuscript. We have addressed the reviewer’s comments below and have given a point-by-point response. We have also attached a revised manuscript in which the amended text has been highlighted using red font for ease of re-review.

Yours sincerely

Clare Collins

On behalf of all authors